Author's response to reviews

Title: Rumination, Anxiety, Depressive Symptoms and Subsequent Depression in Adolescents at Risk for Psychopathology: a Longitudinal Cohort Study

Authors:

Paul O Wilkinson (pow12@cam.ac.uk)
Tim J Croudace (tjc39@cam.ac.uk)
Ian M Goodyer (ig104@cam.ac.uk)

Version: 2 Date: 26 March 2012

Author's response to reviews:

Dear BMC Psychiatry

As requested, I have added extra details to state that both participants and parents gave informed consent to take part in the study, on page 13, para 2.

Yours sincerely

Paul Wilkinson